Author's response to reviews

Title: CCl4 induced genotoxicity and DNA oxidative damages in rats: Hepatoprotective effect of Sonchus arvensis

Authors:

Huda Alkreathy (halkreathy@gmail.com)
rahmat A Khan (rahmatgul_81@yahoo.com)
Muhammad Rashid Khan (mrkhangu@yahoo.com)
sumaira Sahreen (sumairasahreen@gmail.com)

Version: 5 Date: 17 September 2014

Author's response to reviews: see over
Dear Editor

BMC Complementary and Alternative medicine

Dear Sir,

Subject: Response to reviewer comments on: 
\textit{CCL}_4\text{ induced genotoxicity and DNA oxidative damages in rats: Hepatoprotective effect of Sonchus arvensis} 

Dear Sir,

Reference to the comments on above manuscript; Enclosed please find a revised version of our Manuscript.

The manuscript was revised according to editor suggestion edited in the manuscript. All points are explained in detail. Now we hope that it will be suitable for publication in your esteemed Journal.

Regards

Dr. Rahmat Ali Khan
Response to Editor Comments

Abstract:

Sonchus arvensis changed into *Sonchus arvensis*

add labitum changed into *ad libitum*

CCl4 treatment changed into *CCl4 treatment*,

SME may be useful in the prevention of hepatic stress changed into *SME could have potential in the prevention of hepatic stress induced by CCl4.*

**Key words:**

Sonchus arvensis changed into *Sonchus arvensis*

**Conclusion:**

Oxidative stress changed into *oxidative stress induced by CCl4.*